Author's response to reviews

Title: An examination of the health and wellbeing of childless women: A cross-sectional exploratory study in Victoria, Australia

Authors:

Melissa L Graham (melissa.graham@deakin.edu.au)
Erin Hill (erin.hill@deakin.edu.au)
Julia M Shelley (julia.shelley@deakin.edu.au)
Ann R Taket (ann.taket@deakin.edu.au)

Version: 4 Date: 26 October 2011

Author's response to reviews:

To Whom it may Concern,

Thank you for accepting for publication our manuscript “An examination of the health and wellbeing of childless women: A cross-sectional exploratory study in Victoria, Australia”.

The following changes have been made to the manuscript:
•The author list has been amended to include middle initials in line with the submission system format
•Table cells have been unmerged
•Highlighted text has been removed
•Typographical, spelling and grammatical errors have been corrected.

Kind regards
Melissa Graham
on behalf of the research team